Author’s response to reviews

Title: Analysis of Total Knee Arthroplasty revision causes in a tertiary care hospital in Germany

Authors:

Anne Postler (anne.postler@uniklinikum-dresden.de)
Cornelia Lützner (cornelia.luetzner@uniklinikum-dresden.de)
Franziska Beyer (franziska.beyer@uniklinikum-dresden.de)
Eric Tille (eric.tille@uniklinikum-dresden.de)
Jörg Lützner (joerg.luetzner@uniklinikum-dresden.de)

Version: 2 Date: 30 Jan 2018

Author’s response to reviews:

Thank you for reviewing our paper again. In detail we have done the following changes.

Editor:

1) You should modify the title. A suggestion would be to remove “in a tertiary care hospital in Germany”
   o We modified the title as you suggested.

2) You need to reformulate your conclusion. Remove the first sentence about tertiary care center and improve the second sentence.
   o We reformulated the conclusion.

Reviewer 1:

1) You did correct the values of table 1, the sums are incorrect on line 1 e 2.
   o We thoroughly checked table 1 and didn’t find any errors. However, we found some mistakes in the first two lines of table 2. We apologize that! The values have been corrected.
2) … correct the references to Vancouver style.

- We changed the style of the references in Vancouver style.